ANSWERING REVIEWERS

Name of journal: World Journal of Cardiology
Manuscript NO: 66000
Title:

Reply to reviewers’ comments:
Table 1 should indicate the reference tissue (eg myocardium) against which the intensities of the lesions are compared. In general, the authors have compiled and edited the subject at a sufficient and good level.

Thank you for your comment. Following your suggestion we modified the table title as follow: “Cardiovascular magnetic resonance characteristic (location, signal intensity and contrast contrast-enhanced relative to that of adjacent normal myocardium) of benign and malignant cardiac masses”